Author's response to reviews

Title: Implementation of secondary fracture prevention services after hip fracture: a qualitative study using extended Normalization Process Theory

Authors:

Sarah Drew (sarah.drew@ndorms.ox.ac.uk)
Andrew Judge (andrew.judge@ndorms.ox.ac.uk)
Carl May (c.r.may@soton.ac.uk)
Andrew Farmer (andrew.farmer@phc.ox.ac.uk)
Cyrus Cooper (cc@mrc.soton.ac.uk)
Kassim Javaid (kassim.javaid@ndorms.ox.ac.uk)
Rachael Gooberman-Hill (R.Gooberman-Hill@bristol.ac.uk)

Version: 5 Date: 1 April 2015

Author's response to reviews: see over
Author's response to Editor

Title: Implementation of secondary fracture prevention services after hip fracture: a qualitative study using extended Normalization Process Theory

Authors:

Sarah Drew\textsuperscript{1}, Qualitative Research Associate
Andrew Judge\textsuperscript{1,2}, Senior Statistician and Associate Professor
Carl May\textsuperscript{3}, Professor of Healthcare Innovation
Andrew Farmer\textsuperscript{4}, Professor of General Practice
Cyrus Cooper\textsuperscript{1,2}, Director and Professor of Rheumatology
M Kassim Javaid\textsuperscript{1,2}, Lecturer in Metabolic Bone Disease
Rachael Gooberman-Hill\textsuperscript{5}, Reader in Applied Health Research
And The REFReSH study group

\textsuperscript{1}Oxford NIHR Musculoskeletal Biomedical Research Unit, Nuffield Department of Orthopaedics, Rheumatology and Musculoskeletal Sciences, University of Oxford
\textsuperscript{2}MRC Lifecourse Epidemiology Unit, University of Southampton
\textsuperscript{3}Faculty of Health Sciences, University of Southampton
\textsuperscript{4}Nuffield Department of Primary Care Health Sciences
\textsuperscript{5}School of Clinical Sciences, University of Bristol

Version: 5 Date: 1\textsuperscript{st} April 2015

Author's response to reviews: see over
Dear Editor,

We are grateful for the Editor’s comments. We address the comments in our revised manuscript, and provide a point-by-point response to each of the concerns. We hope that you will find the revisions satisfactory.

On behalf of all authors,

Sincerely yours,

Sarah Drew
Editorial note: Prof Anne Elizabeth Rogers

We thank the Editor for their comment. On the basis of this comment we have made the following change to the manuscript.

1. Happy to accept this for publication now but the references will need to conform to journal style before it goes through the rest of the editing and acceptance process

Author response and action:
Thank you for this comment, we apologise for this error. We have re-formatted the references and sincerely hope that the manuscript is now ready for publication.